Author's response to reviews

Title: Use of traditional cooking fuels and the risk of young adult cataract in rural Bangladesh: a hospital-based case-control study

Authors:

Joydhan Tanchangya Dr (tanchangya81@yahoo.com)
Alan Frederick Geater Dr (alan.g@psu.ac.th)

Version: 2 Date: 7 November 2010

Author's response to reviews: see over
Ref: Submission of revised manuscript, MS: 1777499463393993 - Use of traditional cooking fuels and the risk of young adult cataract in rural Bangladesh: a hospital-based case-control study.

Dear Editors,

BMC Ophthalmology.

We would like to submit the revised manuscript of the research article entitled “Use of traditional cooking fuels and the risk of young adult cataract in rural Bangladesh: a hospital-based case-control study” to your journal. Our detailed responses to the comments by the three reviewers are also included. We apologize for the long delay in submitting this revision.

As previously stated in our original submission, the findings of this research have not been submitted or published in any other journals or other citation forms. In addition, all authors of this study have read and agreed to its contents, including the latest revisions. Ethical approval for the study was granted by the Ethics Committee of the Faculty of Medicine, Prince of Songkla University, Thailand, and by the management of Impact Foundation Bangladesh. Documented informed consent was also provided by all respondents.

We hope that the revised manuscript has adequately addressed all comments by the reviewers and will be acceptable for publication in your journal.

Sincerely yours,

Joydhan Tanchangya
Corresponding author
tanchangya81@yahoo.com